Author's response to reviews

Title: Magnesium supplementation and high volume hydration reduce the renal toxicity caused by cisplatin-based chemotherapy in lung cancer patients; a toxicity study.

Authors:

Takako Oka (takako_mix@yahoo.co.jp)
Tatsuo Kimura (kimutats@med.osaka-cu.ac.jp)
Tomohiro Suzumura (tsuzumura@msic.med.osaka-cu.ac.jp)
Naoki Yoshimoto (yoshimoto_fuucyan@yahoo.co.jp)
Toshiyuki Nakai (shintokumaru1984@mail.goo.ne.jp)
Norio Yamamoto (norisuke-1020@softbank.ne.jp)
Kuniomi Matsuura (kuniomi@med.osaka-cu.ac.jp)
Shigeki Mitsuoka (micchan@msic.med.osaka-cu.ac.jp)
Naruo Yoshimura (y-naruo@sc4.so-net.ne.jp)
Shinzoh Kudoh (shinzohykudoh@med.osaka-cu.ac.jp)
Kazuto Hirata (kazutoh@msic.med.osaka-cu.ac.jp)

Version: 7
Date: 14 May 2014

Author's response to reviews:

Thank you very much for your third time advices that the paper needed Trial Registration Number.

We added the UMIN number at the last of the abstract.

Please let us know if you have any additional suggestions or concerns.